
Present : Joel Closier, Andrew Smith, Marcos Seco Miguelez, Gianluca Castellani, Roberto Santinelli, Raja 
Nandakumar, Gennady Kuznetsov, Cedric Poterat, Ricardo Graciani-Diaz, Marianne Bargiotti, Stuart 
Paterson, Nick Brook, Andrei Tsaregorotsev, Roland Bernet, Umberto Marconi

Simulation: 
nothing to report. Nick will check when we will do the di-muon production. It will be done when the new 
Gauss version will be ready.

Reconstruction:
Cern: lost request are in the queue. We have again the same problem that one or severals tapes have 
requests which are stuck somewhere and which are blocking lot of jobs. To be investigate with IT 
people.
RAL: problem with some tape. Entries will be remove from the LFC and transfer will be issued. Files 
which will be unavailable after this operation will be remove from the processing DB.
Gridka : mappin g is fixed. We have to retry.

Problem to submit jobs from the processing DB to a target site.

What to do when a job is failing at a given site because one or several files are unavailable.  This has to 
be discussed in the PASTE meeting. Joel should be prepare some use case..

Stripping:
Check the ETC and why the DST are cut after 1.9 GB...
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